Author’s response to reviews

Title: Primary glioblastoma in the pineal region: a case report and review of the literature

Authors:

Kyung-Sub Moon (moonks@chonnam.ac.kr)
Shin Jung (sjung@chonnam.ac.kr)
Tae-Young Jung (tongbori@hanmail.net)
In-Young Kim (kiy87@hanmail.net)
Min-Cheol Lee (mclee@chonnam.ac.kr)
Kyung-Hwa Lee (azimmed@hanmail.net)

Version: 2 Date: 14 February 2008

Author’s response to reviews: see over
Author's covering letter for initial submission

Title: Primary glioblastoma in the pineal region: a case report and review of the literature

Authors:

Version: 1 Date: 15 January 2008

Comments:

Michael Kidd, Jan 15, 2008
Editor-in-Chief, Journal of Medical Case Reports

Dear Professor Michael Kidd

We are enclosing the manuscript entitled, Primary glioblastoma in the pineal region: a case report and review of the literature, for consideration of publication in Journal of Medical Case Reports.

Until now, only seventeen cases of glioblastoma in the pineal region have been described in the literature. We discuss its clinical courses, radiological findings, and treatment strategies with a review of the relevant literature.

The material contained in the manuscript has not been published previously, nor is it under consideration for publication elsewhere. All authors have read and approved the submission of this manuscript. We hope that you will find this work acceptable to Journal of Medical Case Reports.

Sincerely yours,

Shin Jung, M.D., Ph.D.
Department of Neurosurgery, Brain Tumor Clinic & Gamma Knife Center, Chonnam National University Hwasun Hospital and Medical School, 160 Ilsim-ri, Hwasun-eup, Hwasun-gun, Jeollanam-do, 519-809, South Korea
Tel:+82-61-379-7666, Fax:+82-61-379-7673, e-mail: sjung@chonnam.ac.kr